Author’s response to reviews

Title: An unusually large benign tumour (myofibroblastoma) of male breast: a case report

Authors:

AMS Abeysekara (abey@hotmail.co.uk)
HP Priyantha HPP Siriwardana (hppsir@hotmail.com)
KF Abbas (kfabbas@hotmail.co.uk)
Peter P Tanner (peter.tanner@bhrhospitals.nhs.uk)
Akin AA Ojo (akin.ojo@bhrhospitals.nhs.uk)

Version: 2 Date: 24 August 2007

Author’s response to reviews:

AMS Abeysekara MSc FRCS
Department of Surgery
King George Hospital
Barley Lane
Goodmayes Essex IG3 8YB
UK.
Tel: 00 44 208 970 8174
Fax: 00 44 208 970 8124
E-mail: abey@hotmail.co.uk

Editor
J Med Case rep

Dear Sir

We submit the manuscript “an unusually large benign tumour (myofibroblastoma) of male breast: a case report” for publication in the Journal of Medical Report. We confirm that the work is original and all authors have contributed actively to the final version of the manuscript, which all authors have seen and approved. We
also confirm that it is not under consideration for publication in any other journal.

This report describes an unusual presentation of a disease i.e. myofibroblastoma of breast, and it is the largest myofibroblastoma reported and we think it is suitable for publication in J Med Case Rep.

Thanking you

Yours faithfully

AMS Abeysekara MSc FRCS

Corresponding author